Author’s response to reviews

Title: Lateral sacral meningocele presentation as a gluteal Mass: a case report

Authors:

Afsoun Seddighi (invincible19152@yahoo.com)
Amir Saied Seddighi (invincible19152@gmail.com)

Version: 6 Date: 15 May 2009

Author’s response to reviews: see over
Dear Editor
Thanks a lot for your guidelines to correct the format of our article. As your order we corrected the grammatical writing errors and reformatted the figures to higher resolution. We provided a better pathologic slide of the lesion and We deleted the name of the patient from the figures. It would be a great honor if you accept this paper in your respectful journal.

Sincerely, Dr. Afsoun Seddighi
           Dr. Amir Saied Seddighi